Author's response to reviews

Title: Antioxidant-rich dietary intervention for improving asthma control in pregnancies complicated by asthma: study protocol for a randomized controlled trial.

Authors:

Jessica A Grieger Dr (jessica.grieger@adelaide.edu.au)
Lisa G Wood Assoc/Prof (Lisa.Wood@newcastle.edu.au)
Vicki L Clifton Assoc/Prof (vicki.clifton@adelaide.edu.au)

Version: 4 Date: 19 March 2014

Author's response to reviews: see over
3 March, 2014

To the editor: BMC Trials

We would like to re-submit our protocol paper: Antioxidant-rich dietary intervention for improving asthma control in pregnancies complicated by asthma: study protocol for a randomized controlled trial, by Jessica A. Grieger, Lisa G. Wood, and Vicki L. Clifton.

We have revised our paper in accordance with the reviewer and editorial suggestions.

We declare this is original research, has not been previously published and has not been submitted for publication elsewhere while under consideration, and no conflict of interest exists for any author.

Yours sincerely,

Dr Jessica A Grieger (first author)